Author's response to reviews

Title: Renin-angiotensin system gene polymorphisms and high blood pressure in Lithuanian children and adolescents

Authors:

Sandrita Simonyte (sandritos@gmail.com)
Renata Kuciene (renatakuciene@yahoo.com)
Jurate Medzioniene (jurate.medzioniene@lsmuni.lt)
Virginija Dulskiene (virginija.dulskiene@lsmuni.lt)
Vaiva Lesauskaite (vaiva.lesauskaite@lsmuni.lt)

Version: 3 Date: 23 Aug 2017

Author’s response to reviews:

Dear Editor,

We kindly appreciate the revision of our manuscript. We hope that the revised manuscript will be acceptable for publication in your journal. We have enclosed our point-by-point response to the comments raised by the editors.

Regarding overlaps in the methods section, the methodology of the measurements of BP, weight, height, and WC was presented in our previous epidemiological study (3. Dulskiene V, Kuciene R, Medzioniene J, Benetis R. Association between obesity and high blood pressure among Lithuanian adolescents: a cross-sectional study. Ital J Pediatr. 2014;40:102. doi: 10.1186/s13052-014-0102-6) that was cited in this manuscript.

Revisions:

Abstract section, line 28-29, page 2
Abstract section, line 37-39, page 2
Methods section, line 112-116, page 5
Methods section, line 117-118, page 5
Methods section, line 124-126, page 6